

Response to reviewers

Reviewer 1: Mechanisms contributing to poor outcome predictors have been explained.

The clinical importance of predictors of poor outcomes have been included.

Reference to role of inflammation and its role in severity of COVID-19 has been highlighted.

Reviewer 2: The review has been reorganized and language corrected

A diagram depicting the pathophysiology of liver involvement in COVID-19 has been incorporated.